

# **Earth's Future**

### **RESEARCH ARTICLE**

10.1029/2023EF003852

### **Special Section:**

Past and Future of Marine Ecosystems

### **Key Points:**

- Assumptions regarding how temperature drives biological rates strongly influence food web responses to warming
- Community interactions further modify outcomes to warming which complicate efforts to generalize warming responses across food webs
- Uncertainty in food web temperature dependences should be represented in climate change projections

### **Supporting Information:**

Supporting Information may be found in the online version of this article.

### **Correspondence to:**

J. C. P. Reum, Jonathan.Reum@noaa.gov

### Citation:

Reum, J. C. P., Woodworth-Jefcoats, P., Novaglio, C., Forestier, R., Audzijonyte, A., Gårdmark, A., et al. (2024). Temperature-dependence assumptions drive projected responses of diverse sizebased food webs to warming. *Earth's Future*, *12*, e2023EF003852. https://doi. org/10.1029/2023EF003852

Received 31 MAY 2023 Accepted 12 FEB 2024

### **Author Contributions:**

Conceptualization: J. C. P. Reum, P. Woodworth-Jefcoats, C. Novaglio, R. Forestier, A. Audzijonyte, A. Gårdmark, M. Lindmark, J. L. Blanchard

© 2024 The Authors. Earth's Future published by Wiley Periodicals LLC on behalf of American Geophysical Union. This article has been contributed to by U.S. Government employees and their work is in the public domain in the USA. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# **Temperature-Dependence** Assumptions Drive Projected Responses of Diverse Size-Based Food Webs to Warming

J. C. P. Reum<sup>1</sup>, P. Woodworth-Jefcoats<sup>2</sup>, C. Novaglio<sup>3,4</sup>, R. Forestier<sup>3</sup>, A. Audzijonyte<sup>3,4</sup>, A. Gårdmark<sup>5</sup>, M. Lindmark<sup>5</sup>, and J. L. Blanchard<sup>3,4</sup>

<sup>1</sup>NOAA Fisheries Service—Alaska Fisheries Science Center, Seattle, WA, USA, <sup>2</sup>NOAA Fisheries Service—Pacific Island Fisheries Science Center, Honolulu, HI, USA, <sup>3</sup>Institute for Marine and Antarctic Studies, University of Tasmania, Hobart, TAS, Australia, <sup>4</sup>Centre for Marine Socioecology, University of Tasmania, Hobart, TAS, Australia, <sup>5</sup>Department of Aquatic Resources, Swedish University of Agricultural Sciences, Uppsala, Sweden

**Abstract** Food web projections are critical for evaluating potential risks to ecosystems and fisheries under global warming. The temperature dependence of biological processes and regional differences in food web structure are two important sources of uncertainty and variation in climate forced projections of fish communities, but we do not know their magnitude or relative contribution. Here we systematically evaluated a range of different assumptions about temperature-dependence on rates, including size-dependent effects, controlling food intake, metabolism, and non-predation mortality in fishes using species-resolved size spectrum food web models that link individual-level physiological processes to population and community dynamics. We simulated the physiological effect of warming in a range of size-structured food web models calibrated to different marine ecosystems and in simplified trait-based models. Higher food intake in warmed conditions increased total fish biomass, catches, and mean body weight, but these effects were offset by the negative effects of warming on metabolism and mortality, which combined resulted in lower total biomasses and catches for most food webs. These effects were enhanced when warming increased metabolic rates more than food intake, and the outcomes were also sensitive to size dependency of temperature responses. Importantly, these general patterns were not uniform across all food webs-individual functional groups and fish species within food webs responded to warming in different ways depending on their position in the food web and its structure. Hence, caution is warranted when generalizing food web or species outcomes to warming because they are mediated by community interactions. Uncertainty related to temperature dependence and ecological interactions will impact food web projections and should be represented in climate change projections.

**Plain Language Summary** Multi-species models are often used to predict how climate change will affect marine ecosystems, fisheries yields and conservation. However, temperature can affect species in a variety of ways, making predictions challenging. Moreover, warming driven changes in a fish community will also depend on species interactions, potentially amplifying or dampening warmingdriven effects in individual species. We explored how 2°C of warming may impact fish biomasses, yields and sizes using six models developed for different marine food webs and three theoretical models with different assumptions about species interactions. We found that, overall, warming tended to result in lower biomasses and fisheries yields, but slightly larger average fish sizes. Yet, results differed depending on how we modeled temperature effects on species food intake and energy expenditure. Moreover, even under the same assumptions about temperature impacts, models developed for different ecosystems behaved differently showing that species interactions will modify warming effects, making general predictions difficult. To make better predictions on warming impacts and more informed adaptation strategies we urgently need more work to understand how temperature affects individual species and communities. Meanwhile, models predicting ecosystems responses to warming should more clearly account for the uncertainty in temperature effects on fishes.

### 1. Introduction

Global mean sea surface temperatures have warmed  $\sim 1^{\circ}$ C relative to preindustrial levels and an additional warming of  $\sim 1.5-3.5^{\circ}$ C is possible over the next century depending on future greenhouse gas emission rates (Kwiatkowski et al., 2020). Our ability to project the impacts of future climate change on marine food webs and to make relevant adaptive changes, depends significantly on understanding the effect of temperature on physiological and ecological processes. At the cellular level, temperature drives metabolism and other biochemical





Formal analysis: J. C. P. Reum, C. Novaglio, M. Lindmark, J. L. Blanchard Methodology: J. C. P. Reum, C. Novaglio, A. Audzijonyte, M. Lindmark Project administration: J. C. P. Reum Software: R. Forestier Supervision: J. C. P. Reum Validation: J. C. P. Reum Visualization: J. C. P. Reum, R. Forestier Writing - original draft: J. C. P. Reum, P. Woodworth-Jefcoats, C. Novaglio, R. Forestier, A. Audzijonyte, A. Gårdmark, M. Lindmark, J. L. Blanchard Writing - review & editing: J. C. P. Reum, A. Audzijonyte

reaction rates that scale up to influence growth, predation, reproduction, and mortality (Brown et al., 2004; Portner & Peck, 2010). Changes in vital rates, in turn, underpin and depend on community interactions (Gårdmark & Huss, 2020; Ohlberger et al., 2011), and influence population extinction risk and the expansion or contraction of species distributions (Zhang et al., 2017). Warming is projected to increase stratification, leading to a reduced surface nutrient supply and a decrease in primary productivity (Bindoff et al., 2019; Doney et al., 2012). Consequently, models project a decline in global marine biomass (Kwiatkowski et al., 2019; Lotze et al., 2019), but the ultimate outcome of warming on food webs and community structure will also depend in part on the interplay between direct effects of temperature on physiology and indirect effects, transmitted through networks of species interactions (Gilman et al., 2010; Lord et al., 2017).

Given the prevalence of feedbacks and the broad spatial and temporal scales on which food webs operate, models are necessary to project climate impacts on species, evaluate ecosystem-based management under climate change, and assess the potential efficacy of mitigation strategies (Horn et al., 2021). However, food web models vary widely in structure and complexity and emphasize different ecological processes and levels of biological detail depending on the question motivating their development. Consequently, food web projections can range widely among models. In recent global marine food web ensemble studies, end-of-century projections of total animal biomass ranged from no change to a decrease of more than 30% across nine different global models under the high emission scenario (RCP 8.5/SSP5-8.5) (Lotze et al., 2019; Tittensor et al., 2021), and even wider ranges were observed in regional projections (Bryndum-Buchholz et al., 2020). One potential reason for variation in food web projections is the difference in the adopted temperature-dependence assumptions, reflecting uncertainty over the representation of individual-level temperature effects and their scaling to ecosystem-level impacts (Heneghan et al., 2021). Another important source of uncertainty is the effect of interactions among species, as global food web model ensembles typically do not resolve species, but focus on functional or trait groups (e.g., Lotze et al., 2019). To improve projections of global warming impacts on fish communities and ocean food webs we need a better understanding of how different temperature-dependencies alter emergent food web properties and depend on species interactions (Heneghan et al., 2021).

It is typically observed that biological rates in food webs increase exponentially with temperature, following the Van't Hoff-Arrhenius equation (Brown et al., 2004). This assumption is also included in many models. While originally developed to describe temperature effects on chemical reaction rates, the Arrhenius equation is now often applied to many physiological and ecological rates over biologically plausible temperature ranges (Brown et al., 2004; Kooijman, 2000). For example, in physiologically structured food web models, temperaturedependencies have been applied to rates controlling body growth and mortality (Blanchard et al., 2012; Lindmark et al., 2019; Maury, Shin, et al., 2007; Thunell et al., 2021; Zhang et al., 2017). However, biological rates do not all scale with temperature in the same manner. Cross-taxa studies indicate that activation energies associated with parameters controlling animal intake rates (i.e., that control the slope of the relationship between rates and temperature) may be lower than those related to metabolic rates (Englund et al., 2011; Rall et al., 2012; Vucic-Pestic et al., 2011), which suggests that fish feeding rates may increase less strongly with temperature than metabolism. Further, the effects of temperature may also depend on body mass, which means that animals of different weight might have different scaling of intake or metabolic rates (Killen et al., 2010; Lindmark, Ohlberger, & Gårdmark, 2022; Ohlberger et al., 2012). An important consequence is that different life stages may respond differently to warming, which may alter population regulation, stage structure, and species interactions, given that predation in aquatic systems is strongly size-structured (Lindmark et al., 2018, 2019; Thunell et al., 2021). Controlled simulation experiments that systematically test how different individual-level temperature hypotheses propagate through species and food webs have until now only been done in single models (Lindmark, Audzijonyte, et al., 2022; Lindmark et al., 2019; Reum et al., 2020), and we do not know how conclusions from these cases can be generalized across other food web models.

To explore the effect of different size- and temperature-dependence of vital rates on food web responses to warming requires models that account for both intra-specific and population-level processes. This includes the effect of intra-specific variation in body size on individual-level physiological rates and how it propagates to population level rates and species interactions. Size spectra represent the abundance of organisms in a community as a function of individual body size (Sheldon et al., 1972), and dynamic size spectrum models track the flow of biomass up the size spectrum (Andersen, 2019; Blanchard et al., 2017). The models are underpinned by extensive theoretical work and empirical evidence that recognizes body size as the primary trait influencing physiological rates and ecological interactions (Andersen, 2019; Blanchard et al., 2017; Kerr & Dickie, 2001; Peters, 1983;

Sheldon et al., 1972; Trebilco et al., 2013). Importantly, the individual-level processes of size-dependent food consumption, body growth, and mortality are based on bioenergetic principles and are, via intraspecific variation in body size, mechanistically linked to population and community dynamics (Andersen et al., 2016; Blanchard et al., 2017). The framework is therefore well-suited to evaluating individual-level climate stressors on emergent food web properties, such as fish community size structure (Woodworth-Jefcoats et al., 2019) and growth rates (Heneghan et al., 2019), as well as species interactions (Lindmark, Audzijonyte, et al., 2022; Reum et al., 2020). Different versions of size spectrum models have been used to project climate impacts on marine systems and fisheries at both global (e.g., Blanchard et al., 2014; Lefort et al., 2015) and regional scales (e.g., Reum et al., 2020; Woodworth-Jefcoats et al., 2013, 2019). They have also demonstrated that temperature-dependence of rates regulating body growth and mortality increases uncertainty in species long-term biomass projections (Reum et al., 2020), and their effects remain important to resolve as warming outcomes are sensitive to food web structure (Gibert, 2019; Lindmark et al., 2019; Sentis et al., 2014; Thunell et al., 2021).

Here, we systematically evaluated the impact of adopting different individual-level temperature dependence hypotheses for rates controlling fish growth and mortality on emergent food web properties using a diverse set of species-resolved, dynamic size-structured food web models. Specifically, we examined (a) the individual and joint impact of adopting standard exponential temperature dependencies on rates controlling fish body growth and non-predation mortality; (b) the impact of assuming lower activation energies on rates controlling food intake compared to metabolism; (c) the effect of assuming size dependent temperature scaling of food intake and metabolism; and (d) the joint effects of temperature dependences in scenario (b) and (c). These scenarios were also applied to a simplified trait-based size spectrum model (TBM; Andersen et al., 2016) to assess how more realistic resolution of species compares to a generalized TBM community where "species" differ only in their asymptotic size. In TBMs, asymptotic size is considered a master trait that embodies a trade-off between reproductive output and asymptotic size, and theoretical arguments linking the two set predator abundances (Andersen et al., 2016). The multispecies size spectrum models (MSSMs) in contrast were previously developed and calibrated to different regional ecosystems spanning subpolar to tropical latitudes, and included oceanic, continental shelf, and nearshore demersal food webs (Audzijonyte et al., 2023; Blanchard et al., 2014; Lindmark, Audzijonyte, et al., 2022; Novaglio et al., 2022; Reum, Blanchard, et al., 2019; Woodworth-Jefcoats et al., 2019). In MSSMs, species-level detail is richer-parameters controlling growth rates, feeding interactions, and maturation and asymptotic sizes are tuned to improve ecological realism and abundances are adjusted to better match observed levels. Collectively, the MSSMs sample a range of system configurations, but share the same structure regarding the representation of individual energy budgets. Their similarity in modeling framework, yet difference in ecological detail thus provides a robust test of whether the impacts of different assumptions regarding temperature-dependence of biological rates are generalizable across systems.

### 2. Materials and Methods

We evaluated the implications of adopting different temperature dependence assumptions on rates governing body growth and mortality in three different versions of a general TBM and in six previously developed regional MSSMs (Table 1). The three TBMs differed by just one key parameter—the predator to prey body mass ratio—which strongly influences trophodynamics (Andersen & Pedersen, 2010; Jennings & Blanchard, 2004; Zhang et al., 2017). The MSSMs, while structurally similar to TBMs, represent additional species-specific detail regarding life history traits, energy budget parameters, and prey species and size preferences, which improves their ability to represent differences in community structure between ecosystems (Andersen et al., 2016; Blanchard et al., 2017; Reum, Blanchard, et al., 2019). MSSMs and TBMs were used to evaluate the extent to which the food web structure may alter warming outcomes under the different temperature dependence assumptions. We first outline the temperature dependency assumptions and follow with descriptions of the food web modeling framework and the incorporation of temperature dependence on key rates governing food intake, metabolism, and non-predation mortality. Although biological rates may scale unimodally with warming over large temperature ranges (e.g., Delong et al., 2017), we limited the analysis to exponential scaling relationships which remain relevant when species live at temperatures below their tolerance limits and warming increments are small to moderate (Brown et al., 2004; Kooijman, 2000).

AGU
ADVANCING EARTH AND SPACE SCIENCES

Model name	Food web description	Reference	Predators	β	$W_{ m inf}$	Links	Connectance
TBM 500	Generic pelagic food web	Scott et al. (2014)	6	500 [500, 500]	1,845 [6, 10,000]	81	1
TBM 2.5K	Generic pelagic food web	Scott et al. (2014)	6	2,500 [2,500, 2,500]	1,845 [6, 10,000]	81	1
TBM 10K	Generic pelagic food web	Scott et al. (2014)	6	10,000 [10,000, 10,000]	1,845 [6, 10,000]	81	1
Coastal Tasmania	Temperate, nearshore rocky reef	Audzijonyte et al. (2023)	17	835 [50, 3,000]	3,855 [39, 16,012]	174	0.6
SE Australia	Temperate, continental shelf	Novaglio et al. (2022)	19	242 [100, 1,000]	8,520 [3, 32,500]	283	0.78
Central North Pacific	Tropical to subtropical, pelagic ocean basin	Woodworth-Jefcoats et al. (2019)	12	125 [100, 400]	108,072 [8,273, 455,400]	142	66.0
Baltic Sea	Subpolar, pelagic inland sea	Lindmark, Audzijonyte, et al. (2022)	3	808 [426, 1,000]	15,016 [21, 44,903]	6	1
North Sea	Temperate, continental shelf	Blanchard et al. (2014)	12	61,011 [22, 398,849]	7,529 [33, 39,851]	144	1
Eastern Bering Sea	Subpolar, continental shelf	Reum et al. (2020)	15	326 [100, 1,000]	7,509 [79, 37,535]	114	0.51

**Earth's Future** 

# Connectance is links/total possible number of linkages (excluding resource spectra linkages). See Table S1 in Supporting Information S1 for details on additional modifications in model structure relative to Scott et al. (2014).





**Figure 1.** (a) Relative change (%) in temperature-dependent rates predicted by the size-independent exponential (Van't Hoff-Arrhenius) model with activation energy (*E*) values of 0.40 and 0.63 and a reference temperature ( $T_{ref}$ ) of 10°C. For reference, the vertical and horizontal lines indicate  $T_{ref}$  and no change in rates, respectively; filled points indicate change in rates with 2°C of warming relative to  $T_{ref}$ . (b) Relative change (%) in rates under 2°C of warming ( $T_{ref} = 10°C$ ), when temperature scaling is assumed to also depend on body size. The temperature-size interaction parameter *c* is set to -0.006. (c) Schematic overview of size spectrum food web models. Bioenergetic processes described at the individual-level (gray arrows) are coupled to demographic processes (black arrows) that drive population and community size structure. Temperature-dependencies were incorporated into processes labeled in orange. Predators consume suitably sized prey (selected according to a feeding kernel), and assimilated energy is allocated to metabolism, growth, and reproduction. Intake rates, in turn, set predation mortality rates on prey populations. The smallest individuals feed exclusively on the background resource spectrum. Additional loss of individuals result from fishing and non-predation (e.g., disease, senescence) mortality. Reproduction produces individuals that join the population at the initial weight of a given species or functional group. Full model descriptions are provided in Section 2 and Supporting Information S1.

### 2.1. Temperature-Dependence

We incorporated temperature dependence using the exponential Van't Hoff-Ahenius equation, which was modified to represent a potential interactive effect of body mass (Brown et al., 2004; Killen et al., 2010; Lindmark et al., 2019). For a given biological rate B, temperature-corrected values for individuals of mass m at temperature T (K) were calculated as:

$$B(m,T) = B(T_{\text{ref}}) \cdot A(m,T)$$
(1)

where  $T_{ref}$  corresponds to the reference temperature at which the correction scalar A is 1. In turn, A is:

$$A(m,T) = m^{c \cdot (T-T_{\text{ref}})} \cdot e^{\frac{E_a}{k} \left(\frac{1}{T_{\text{ref}}} - \frac{1}{T}\right)}$$
(2)

where k is the Boltzmann constant ( $8.62 \times 10^{-5} \text{ eV K}^{-1}$ ),  $E_a$  is the activation energy (eV), and c is the effect of temperature on the allometric exponent. The exponential term is a modified version of the standard Van't Hoff-Arrhenius temperature correction (Brown et al., 2004) that scales rates relative to the reference temperature (e.g., Maury, Faugeras, et al., 2007). When c is set to 0, the body mass term is unity and the formula reduces to the modified Van't Hoff-Arrhenius correction, with an exponential increase in the rate with temperature that is the same across all body sizes. Different values of c and  $E_a$  result in markedly different temperature scalars across body sizes (Figure 1).

### 2.2. Size Spectrum Models

All models in the comparison were based largely on source code for the R package "mizer" (Scott et al., 2014) which implements TBMs and MSSMs (see https://sizespectrum.org/mizer/ for details). The regional MSSMs included additional modifications from the original source code (Table S1 in Supporting Information S1), but the main equations that compose the models are based on those of Hartvig et al. (2011) as implemented in "R" by Blanchard et al. (2014) and Scott et al. (2014). The models provide predictions of the population dynamics of each *i*th species (or functional group in the case of TBMs) as described by its size spectrum, denoted  $N_i(m)$ , where *m* corresponds to body mass (g). The numerical density of individuals in the body mass range from  $m_1$  to  $m_2$  is calculated from  $\int_{m_1}^{m_2} N_i(m) dm$  and the total biomass density (g m<sup>-3</sup>) of individuals is  $\int_{m_1}^{m_2} N_i(m) mdm$ . The size spectrum is obtained from a numerical solution of the McKendrick–von Foerster equation (Andersen et al., 2016) in which growth is balanced by mortality:

$$\frac{\partial N_i(m)}{\partial t} + \frac{\partial g_i(m) N_i(m)}{\partial m} = -\mu_i(m) N_i(m)$$
(3)

where  $g_i(m)$  is the growth rate (mass per unit time) and  $\mu_i(m)$  is the mortality (per unit time). The individual-level processes of growth and mortality are scaled to the population-level size spectrum following Equation 3, thus removing the need for explicit individual-based simulations (Blanchard et al., 2017). The number of individuals recruiting into the population ( $R_i$ ) is represented as a boundary condition, and individuals enter the population at the smallest population size class,  $m_0$  (here either egg mass, mass at parturition in the case of live-bearing species, or other minimum mass set by the user):

$$g_i(m_0)N_i(m_0) = R_i \tag{4}$$

By default, recruitment into the smallest size group is modeled following a Beverton–Holt type stock–recruit relationship (Scott et al., 2014), though constant recruitment or other stock-recruit relationships can be employed (e.g., Jacobsen et al., 2017; Reum et al., 2020). In addition to representing species or functional groups, the model also includes resource spectra to represent other food sources (Andersen et al., 2016; Blanchard et al., 2009). The model allows for species-specific traits that control growth rates, maturation, egg production, recruitment, and maximum size (Scott et al., 2014).

Growth in Equation 3,  $g_i(m)$ , is calculated from the quantity of ingested prey after accounting for losses owing to assimilation efficiency and metabolism (Hartvig et al., 2011). Metabolism scales with mass allometrically and in most MSSMs it is parameterized to correlate to the maximum food intake rate so that species with higher maximum intake and therefore also potential growth rate also have higher species-specific metabolic rates (Scott et al., 2014). Food intake emerges from the encounter rate of prey and the predator functional response (Scott et al., 2014). Prey encounter rate is the product of the volumetric search rate of predators, prey density, and the predator's prey species and size preferences. The models assume a Holling Type II functional response, and thus capture the effect of predator satiation at high prey encounter rates. Maximum intake rates of the functional response scales with body size allometrically and, where data is available, is derived from species-specific von Bertalanffy growth rate parameters (Scott et al., 2014). In this study we included temperature dependence on metabolic rates and on rates controlling both volumetric search and maximum intake rates, based on earlier studies (Blanchard et al., 2012; Rall et al., 2012), and extensive empirical evidence (Englund et al., 2011; Gillooly et al., 2001; Killen et al., 2010; Rall et al., 2012).

Processes that contribute to mortality in the models include fishing, predation, and non-predation mortality resulting from, for instance, disease or senescence (Andersen et al., 2016; Scott et al., 2014). Non-predation mortality is assumed to depend allometrically on a species maximum body mass, but is constant across body size classes within a species (Scott et al., 2014). The interspecific variation in non-predation mortality reflects expectations from life history theory and empirical findings showing lower mortality rates in larger, longer-lived species compared to small-bodied, short-lived species (Pauly, 1980; Thorson et al., 2017). Imposing constant non-predation mortality across body size classes has the practical benefit of preventing an unrealistic buildup of large-bodied, high trophic level individuals (Andersen et al., 2016), and is perhaps the simplest assumption given the different sources contributing to non-predation mortality and uncertainty with respect to their cumulative

dependency on individual body mass, though other size-dependencies have been explored (e.g., Law et al., 2009). That said, cross-taxa and intra-specific studies show that non-predation mortality and individual lifespans in laboratory and natural populations are temperature-dependent (Keil et al., 2015; Munch & Salinas, 2009). We therefore also incorporated temperature dependence on non-predation mortality rates, assuming higher non-predation mortality at higher temperatures.

### 2.3. Trait-Based and Regional Multispecies Size Spectrum Models

To explore temperature impacts on marine ecosystems, we used three different versions of trait-based models (TBMs), each with a different value of  $\beta$ , the preferred predator to prey body mass ratio parameter (Table 1). The  $\beta$  parameter is an important determinant of food chain length, size spectrum steepness, and the strength of trophic cascades (Andersen & Pedersen, 2010; Jennings & Blanchard, 2004), and could thus also be important for emergent community responses to warming. When  $\beta$  is low, trophic level increases more rapidly with body mass and the slope of the size spectrum steepens (Jennings & Blanchard, 2004). In low  $\beta$  systems, more trophic levels are present in the size spectrum relative to high  $\beta$  systems (Jennings & Blanchard, 2004), which amplifies the downstream and upstream effects of predator or prey resource perturbations (Andersen & Pedersen, 2010). For TBMs we selected  $\beta$  values of 500, 2,500, and 10,000 which roughly span the magnitude of values of community-wide  $\beta$  inferred from stable isotope and stomach content analyses (Al-Habsi et al., 2008; Coghlan et al., 2022; Reum et al., 2015; Reum, Holsman, et al., 2019; Trebilco et al., 2016) and mean community values from the regional MSSMs (Table 1).

All three TBMs include nine functional groups ranging in maximum size from 6 to 10,000 g, all equally spaced on a logarithmic scale (Table 1). Size at the onset of maturation (in grams) was 25% of their maximum size. We assumed that the community consisted of generalists and that all functional groups were able to feed on each other. Specifically, all species interaction matrix values were set to 1 such that all prey are consumed in proportion to their relative abundances after adjusting for the size selectivity of predators (Hartvig et al., 2011; Reum, Blanchard, et al., 2019). All functional groups were fished assuming a fishing mortality of 0.1 yr<sup>-1</sup> and a knife-edge selectivity function, where fishing mortality starts at size-at-maturity. Additional details regarding TBM parameterization are provided in Table 1.

Next, we explored temperature effects on community dynamics using six previously developed regional multispecies size spectrum models (MSSMs; Table 1). These models were parameterized to represent ecologically and economically important species composing food webs from the eastern Bering Sea shelf (Reum, Blanchard, et al., 2019), North Sea (Blanchard et al., 2014), Baltic Sea (Lindmark, Audzijonyte, et al., 2022), central North Pacific Ocean (Woodworth-Jefcoats et al., 2019), Southeast Australian shelf and slope (Novaglio et al., 2022), and coastal Tasmania, Australia (Audzijonyte et al., 2023; Table 1). In each MSSM, an effort was made to include species-specific parameters controlling life history, growth rates, fishing mortality, and feeding ecology in terms of the prey and species preference of predators. Overall, the amount of species-specific detail, inclusion of further model refinements, and approach to model calibration and validation (see Table S1 in Supporting Information S1) varied across systems based on data availability and the original ecological or management issue motivating the model. Additional details on distinctions between models and summaries of modifications to model structure relative to Scott et al. (2014) are provided in Table S1 of the Supporting Information S1 and in key references for each system (Table 1). In this study, these regional models were further modified to include the temperaturedependencies described above. No other temperature dependencies or species-specific thermal tolerances were used in the models, and the carrying capacity and growth rate of background resource prey was held fixed to simplify and improve interpretability of the comparisons.

### 2.4. Temperature Dependence Simulation

We conducted two sets of simulation experiments in which the impacts of temperature dependence assumptions were compared to outcomes under a baseline (non-warming) scenario. Following Equation 2, rates are all scaled relative to the reference temperature for a food web. As the reference temperature increases, the same warming increment results in larger relative correction scalars. However, applying region-specific reference temperatures would weaken our ability to distinguish the effects of food web structure on warming outcomes because model structure would be confounded by reference temperatures. To facilitate comparison between models from different biogeographic regions, we therefore defined a common reference temperature of 10°C and evaluated the



### Table 2

Overview of Parameters for Activation Energy  $E_a$  and Mass Exponent c (See Equation 2) Parameter Values Applied to Food Intake, Metabolism, and Non-Predation Mortality Rates

		Parameter values						
			с					
Temperature-dependence	Assumption for rates (code)	Intake	Metabolism	Intake	Metabolism	Mortality		
Van't Hoff-Arrhenius	Baseline (Base)	0	0	0	0	0		
	Intake (Int)	0	0	0.63	0	0		
	Metabolism (Met)	0	0	0	0.63	0		
	Mortality (Mort)	0	0	0	0	0.63		
	All (All)	0	0	0.63	0.63	0.63		
Extensions	Non-uniform scaling (All_E)	0	0	0.4	0.63	0.63		
	Allometric scaling (All_c)	-0.006	-0.006	0.63	0.63	0.63		
	Combination (All_cE)	-0.006	-0.006	0.4	0.63	0.63		

*Note.* For all temperature-dependence assumptions and models, the reference temperature ( $T_{ref}$ ) was set to 10°C. The baseline simulations included no warming relative to  $T_{ref}$ ; the remaining simulations with different temperature-dependence assumptions included 2°C of warming.

effects of  $2^{\circ}$ C of warming in each of the modeled ecosystems. In doing so, the same relative change in physiological and mortality rates was applied across all systems under a given temperature dependence assumption. The warming increment of  $2^{\circ}$ C approximates end-of-century global mean sea surface conditions relative to current levels under a future high greenhouse gas emissions scenario (SSP3; Kwiatkowski et al., 2020). For the baseline simulations, the environmental temperature was set equal to the reference temperature, which yields temperature scalars of 1 (see Equation 2).

In the first set of simulation experiments, we applied identical size-independent exponential temperaturedependence to all rates as commonly implemented in food web models (e.g., Blanchard et al., 2012; Maury, Faugeras, et al., 2007; Maury, Shin, et al., 2007; Reum et al., 2020), and evaluated the separate and combined effect of including temperature dependence on food intake, metabolism, and non-predation mortality (Van't Hoff-Arrhenius scenarios in Table 2). This also acted as a sensitivity test to evaluate the effects of turning on and off temperature dependence on one rate at the time. We used an activation energy ( $E_a$ ) of 0.63 eV for each rate. The value has been adopted previously (e.g., Blanchard et al., 2012; Jennings et al., 2008; Reum et al., 2020) and was obtained from a cross-taxa average of  $E_a$  estimates associated with population growth (Savage et al., 2004). For prey intake rates, we assumed temperature-dependencies on volumetric search and maximum intake rates were strongly interlinked and varied with temperature identically (see Section 2.2 for rationale).

The second set of assumptions include temperature-dependencies supported by more recent evidence (Englund et al., 2011; Lindmark, Ohlberger, & Gårdmark, 2022; Rall et al., 2012), but have yet to be widely adopted. Namely, we conducted simulations where  $E_a$  varied among rates and where rates scaled interactively with size and temperature for intake and metabolism (Extension scenarios in Table 2). Specifically, values of  $E_a$  associated with intake were set to 0.40, reflecting lower activation energies (compared to metabolic temperature sensitivities) reported in experiments (Rall et al., 2012). Values for the temperature-size interaction parameter c were set to -0.006 based on a cross-species average of values for metabolic rates in fishes (Killen et al., 2010). We assumed that, if present, the same temperature-size interaction (c) would apply to both metabolism and feeding rates (search rate and maximum intake rate). This assumption was necessary because experimentally measured values of c are scarce for feeding rates, but deemed reasonable because metabolism drives other biological and ecological rates (Brown et al., 2004) and similar allometric temperature dependence is required for both metabolism and feeding rates to achieve realistic scaling of optimum growth rates with mass (e.g., García et al., 2011).

For each set of temperature assumptions, models were initialized using population size spectra estimated from survey data, stock assessments, or according to analytical solutions for theoretical population size structure at equilibrium (Andersen et al., 2016) and iterated forward for 100 years or until equilibrium or quasi-equilibrium conditions were reached. If present, the effects of oscillatory dynamics were removed by averaging simulation

outputs over 20 years which was sufficient to stabilize mean values and facilitate comparisons between model results.

### 2.5. Quantifying Food Web Responses and Coherence

For both sets of temperature assumptions, we compared differences in equilibrium total community biomass, spawning stock biomass (SSB), catches, and mean body size between temperature assumption scenarios and baseline outcomes. The same variables (biomasses, mean size, catches) were also evaluated at the species or functional group level to understand their contribution to aggregate community patterns.

To visualize the similarity with which different systems responded to warming across temperature assumptions, we performed ordination on community-level outcomes (total SSB, catches, and mean body mass) using nonmetric multidimensional scaling (NMDS; Legendre & Legendre 2012). Outcomes were expressed in terms of percent change relative to the baseline scenario and dissimilarity was calculated as Euclidean distance and submitted to NMDS. Using the same approach, we also visualized similarity of the different temperature assumptions based on outcomes across the different food webs. In the resulting NMDS ordination plots, proximity in ordination space corresponds to similarity in model responses.

To assess the level of qualitative similarity (response direction) in model outputs under the temperaturedependence assumptions, we calculated sign agreement in outcomes for community-level variables (e.g., Lotze et al., 2019; Reum et al., 2020). Outcomes were again expressed in terms of percent change relative to the baseline scenario. Sign agreement for a given variable was calculated following  $100 \times |P - N|/(P + N)$ , where Pand N are the number of models in which the relative change was positive or negative, respectively. If half of the models produce positive outcomes and the other half produce negative outcomes, sign agreement is 0% because every positive outcome is matched by an opposite negative outcome. A value of 100% indicates all outcomes change in the same direction. Within a given model and temperature-dependence assumption, we also evaluated uniformity in the sign responses of individual functional groups or species. This was accomplished by calculating sign agreement across species or functional groups.

### 3. Results

### 3.1. Temperature-Dependence Assumptions

The response of total biomass, total SSB, total catches, and mean community body mass to warming was strongly driven by the temperature dependence assumption, with the largest differences across assumptions emerging in the TBMs, and more variable responses in regional MSSMs (Figures 2a and 2b). In all cases, total biomass was closely correlated with SSB; we therefore limit results to SSB to avoid redundancy. Temperature dependence assumptions led to a continuum of responses, depending on how they affected individuals' energy budgets. At one extreme, if warming only led to increased food intake (assumption "Intake"), then biomasses, catches and mean sizes generally increased in most systems (except for catches for the EBS model; Figure 2b; Table 3). The increase in some cases was quite large (>15%), setting the outcomes under this temperature assumption apart from others in ordination space (Figure 2a). Such increases indicate that, at a minimum, abundant species in the systems were not limited by food availability; higher search rates and maximum feeding rates led to higher emergent food intake ("Metabolism"), energy available for growth decreased and therefore lowered biomasses, catches and mean sizes in a majority of systems (Figure 2b; Table 3). If temperature only affected non-predation mortality ("Mortality"), responses were also generally negative particularly for biomasses and catches, but not as strong as when only metabolism was affected (Figure 2b; Table 3).

Interestingly, when temperature affected all four rates in the same way ("All"), biomass and catches in TBMs still decreased compared to the baseline simulations, but mean body sizes were relatively unchanged (Figure 2b; Table 3). On average, regional models responded similarly in terms of biomasses and catches (ca. 2%–3% decrease), and mean body mass (ca. 6% increase) but responses were uniform across models only for mean body mass (Figure 2b; Table 3). When temperature affected all rates but metabolic rates increased faster than food intake rates ("All + Ea"), biomasses and catches decreased even further (5%–8% decrease), and the increase in mean body sizes was considerably smaller (0.6% increase). In this case a metabolic mismatch occurred—growth slowed and total SSB and total catches tended to decline with warming relative to the baseline scenario, and all





**Figure 2.** (a) Left column: NMDS ordination plots depicting similarity of temperature dependence assumption (see Table 2 for codes) across systems based on community-wide outcomes for total spawning stock biomass, total catches, and mean body weight; distance between points conveys relative differences in outcomes under the different temperature dependence assumptions averaged across models; open circles: temperature dependence assumption simulations for individual rates; closed circles: simulations with temperature dependence assumptions on all rates. Right column: relative similarity of model outcomes across temperature assumptions. Models that responded more similarly across temperature dependence assumptions are closer in ordinate space. Trait based model and regional MSSMs are demarcated by yellow and gray convex hulls, respectively. (b) Relative change in equilibrium levels of community total spawning stock biomass, catches, and mean body mass with 2°C of warming assuming temperature dependence assumptions on individual rates (open circle) and all rates (open circles; Table 2). Horizontal dashed line corresponds to 0% (no change). Values for mean weight for the trait based models under the Intake scenario are outside the *y*-axis range and are indicated in the figure.

three responses (including mean body size) decreased relative to the case when all rates were affected by temperature in the same way (Figures 2a and 2b; Table 3).

Next, we investigated outcomes of temperature dependence assumptions where activation energies for all physiological rates were the same, but the effect of warming on physiological rates was stronger among small-relative to large-bodied individuals ("All + c"). Notably, under this assumption, warming generally led to the smallest changes in total SSB and total catches and a large increase in mean body weight (Figures 2a and 2b; Table 3). These changes tended to be smaller on average compared to when all rates had the same activation energies and no size dependency. Yet, if size dependency was combined with the metabolic mismatch assumption where metabolism increased faster than intake ("All + cEa"), impacts on food webs were more similar to those observed under the "All" assumption (Figures 2a and 2b). This is because the size dependency of the physiological rate increases meant that food intake increased relatively faster in small individuals under warming, but the increase in potential growth was offset by a relatively faster overall increase in metabolic rates (Figures 2a and 2b).

### Table 3

Model Averaged Percent Change (and % Sign Agreement Across Models) of Community Total Spawner Stock Biomass (SSB), Catches, and Mean Body Weight

	% Change SSB		% Change	e catches	% Change mean weight	
Temperature-dependence assumption	TBM	Regional	TBM	Regional	TBM	Regional
Intake	18.4 (100)	13 (100)	16.6 (100)	6.6 (66.7)	82.6 (100)	15.9 (100)
Metabolism	-19.7 (100)	-13.2 (100)	-19.1 (100)	-9.2 (66.7)	-29.3 (100)	-2.9 (33.3)
Mortality	-5.5 (100)	-6.2 (100)	-5 (100)	-3.3 (100)	-21.3 (100)	-1 (0)
All	-5.1 (100)	-3.2 (66.7)	-5 (100)	-1.8 (0)	0.2 (33.3)	6.4 (100)
All_E	-12 (100)	-8.5 (66.7)	-11.5 (100)	-5.3 (66.7)	-17.6 (100)	0.6 (33.3)
All_c	2 (100)	-0.2 (33.3)	2.4 (100)	1.3 (33.3)	-4.9 (100)	9.7 (100)
All_Ec	-5.1 (100)	-6 (100)	-4.5 (100)	-2.9 (66.7)	-20.7 (100)	3.3 (66.7)

*Note*. Outcomes for temperature-dependence assumptions with inconsistent directional responses (% sign agreement <100%) are in bold.



**Figure 3.** Inter-model variation (expressed as the standard deviation, SD) of relative responses (% change) under warming within individual temperature assumptions (light gray circles) and inter-temperature assumption variation of responses under warming within individual models (gray triangles) for the four simulations where all rates were temperature dependent (All, All + *E*, All + *c*, and All + Ec). Averages of the SDs for inter-model (black circle) and inter-temperature assumptions (black triangle) are provided.

### 3.2. Variation Across Food Webs

In the three TBMs, the direction of change in total SSB, total catches, and mean body mass were similar under nearly all temperature-dependence assumptions (Table 3), indicating that the large differences in the predator prey mass ratio assumed in these models did not have a major effect on the aggregate food web properties. Averaged across the four main temperature dependence assumptions (All, All + *E*, All + *c*, and All + Ec), inter-model variation for SSB and catches were low (standard deviation: 2.3% and 1.9%, respectively) but somewhat higher for mean body mass (ca. 6.4; Figure 3). In contrast, variation in responses between temperature assumptions, averaged across the three TBMs models, was over twice as high for all response variables (ca. 5.6%, 5.5%, and 12.5%, respectively; Figure 3). This similarity of responses across the TMBs also extended to the biomasses and mean sizes of individual functional groups within the TBMs, which responded to warming in a nonuniform (Table S2 in Supporting Information S1) but highly size-structured manner (Figure 4), reflecting emergence of strong size-based trophic cascades.

In the regional MSSMs, the average direction of change observed for total SSB and catches were largely similar to those for the TBMs under the different temperature dependence assumptions (Figure 2b, Table 3). However, agreement between individual models was lower for most temperature dependence assumptions (Table 3), tempering simple generalizations of the net directional effect of warming on aggregate food web properties in these more ecologically detailed models. Outcomes for mean body mass differed more from those observed for the TBMs: on average, body weight increased when temperature dependence was applied to all four rates, though model agreement was mixed when adopting a lower activation energy value (Table 3). Overall, inter-model variation for SSB and catches within the temperature dependence assumptions was higher than that observed for the TBMs, but not for mean body mass (Figure 3). Further, in MSSMs intermodel variation was more similar in magnitude to the inter-temperature variation, that is, variation in responses across temperature dependence assumptions (Figure 3). Species biomass and body mass responses to warming under the regional models were also not uniform in all cases (Table S1 in Supporting Information S1) but lacked the same degree of size-structuring as in the TBMs (Figure 4), reflecting heterogeneity in both the species and size preference of predators.



## **Earth's Future**



Figure 4. Relative change in functional group or species spawning stock biomass (top panels) and mean body mass (bottom panels) under  $2^{\circ}$ C of warming and under different temperature-dependence assumptions (columns). Responses are depicted against the maximum body mass of functional groups and species composing the models. Convex hulls demarcate species responses within a food web model. The dashed line corresponds to 0 (no change).

A key distinction between the TBM and regional MSSMs is the intensity of predation pressure (Figure S1 in Supporting Information S1). In TBMs, predation rates are higher than in the regional models across most size classes (Figure S1 in Supporting Information S1), reflecting lower dependence of functional groups on resource spectra for food and stronger trophic interactions between functional groups. Lower predation mortality rates in the regional models is partly explained by the more abundant resources, which also extend into larger size classes in these models. In regional MSSMs, this is often done to account for larger-bodied prey that are not resolved at species- or functional group level. Consequently, warming driven changes in physiological, and especially intake rates in the TBMs are amplified because of stronger feedbacks between the availability of prey and the growth rate of predators; this is clearly seen when only intake is assumed to depend on temperature (Figures 2a and 2b).

### 4. Discussion

Our comparison of temperature-dependence assumptions across different models yielded at least three important insights. First, we show that the net outcome of warming in food webs is partly dependent on the balance of warming effects on opposing processes that regulate body growth and predation. Ultimately, the direction of the response of aggregate system- and species-level variables will depend on their relative balance under warming.

Second, we show that warming may reduce biomass and catches, but adopting different plausible assumptions regarding activation energy, size-dependent temperature scaling, and their joint occurrence can impact the magnitude and, in some models, the direction of variable responses. Given uncertainty in how the temperature-dependencies should best be captured in ecosystem models, this ambiguity should be acknowledged and incorporated into climate change projections of food webs. Lastly, despite sharing a common size- and physiologically based modeling framework and identical temperature scenarios, we show that the magnitude of variation among temperature assumptions was similar to variation among regional food web models, making simple generalizations on how warming affects fish communities challenging. Rather, our comparisons support the observation that system responses to global warming may require detailed knowledge of species interactions —in particular when considering the differences between the trait-based models and the food web models calibrated to fit a specific system (e.g., Gilman et al., 2010).

Our analysis highlights how adopting temperature dependence assumptions that result in metabolic mismatches or represent size-dependencies in scaling of rates of biological processes are consequential for projecting global warming impacts on food webs. Food web models, including various versions of size spectra models (Blanchard et al., 2012; Jennings et al., 2008; Maury, Faugeras, et al., 2007; Maury, Shin, et al., 2007; Reum et al., 2020), have typically applied the same von Hoff's-Arrhenius temperature corrections to maximum intake, metabolism, and mortality following the metabolic theory of ecology (Brown et al., 2004; Gillooly et al., 2001). However, a metabolic mismatch has been observed in meta-analyses of experimental data (Lemoine & Burkepile, 2012; Rall et al., 2012), and when implemented in size-structured food webs, total SSB, catches, and mean body size generally decreased relative to when uniform activation energies were applied across rates. Moreover, when applying negative body size-dependent scaling, a phenomenon also supported by experimental evidence (Beamish, 1964; García et al., 2011; Messmer et al., 2017; Ohlberger et al., 2012; Strong & Daborn, 1980; Xie & Sun, 1990), the response variables increased slightly. Combined, both measures partially offset each other and system-level variables were more similar to outcomes based on uniform scaling, but importantly, this did not necessarily reflect similarity in outcomes for individual species. Given current uncertainty in how to best represent temperature-dependencies in natural food webs (Lindmark, Audzijonyte, et al., 2022), we suggest analysts acknowledge the uncertainty and carry it through into projections using ensemble modeling frameworks if feasible (Cheung et al., 2016; Gårdmark et al., 2013; Spence et al., 2018). That said, the temperature assumptions we evaluated are not exhaustive and their universality requires further testing. For instance, metabolic mismatch (i.e., faster increase in metabolic rates vs. intake rates) was not supported in an inter-generational experimental study that showed that after three generations baseline metabolic rates of fishes kept at 26 and 30°C remained unchanged, but intake rates were higher at elevated temperatures (Wootton et al., 2022). As empirical evidence accumulates and theory develops, we anticipate that the plausible set of alternative temperature dependence assumptions will also shift.

The use of multiple food web models to evaluate the effects of different temperature dependence assumptions provides a new perspective to previous studies that have explored sensitivities to assumptions using only a single food web (e.g., Lindmark, Audzijonyte, et al., 2022; Reum et al., 2020). Importantly, by using multiple different regional MSSMs but identical temperature dependence assumptions and temperature increase, we show that variability in the magnitude of outcomes of system-level responses across the different assumptions is comparable to variability across food webs. Further, under a subset of assumptions and model combinations, the directional response of system-level variables were inconsistent across food webs; modeled food webs that respond most negatively under warming tended to have higher predation levels and lower reliance on basal resources. Specieslevel responses, particularly SSB, were also mixed within most food webs and changed across temperaturedependence assumptions. This heterogeneity highlights the importance of species-level interactions in mediating population- and system-level responses to climate change, as also discussed elsewhere (Gårdmark & Huss, 2020; Gilman et al., 2010; Lotze et al., 2019; Tekwa et al., 2022). It also suggests that global food web models that typically consist of simplified, generic functional groups (e.g., Petrik et al., 2019) may have more limited predictive capacity at regional scales than currently assumed. This point can be seen when comparing trait based (TBMs) and multi-species (MSSMs) model outcomes in our study. For biomass and catches the overall direction of responses was often similar across temperature scenarios, but the magnitude differed. In contrast, for mean body sizes TBMs often predicted a decrease with warming (in more complex temperature scenarios), while multi-species regional models overall suggested an increase.

We intentionally limited our temperature-dependency assumption to those associated with individual-level rates impacting body growth and mortality to identify how changing processes at that scale propagate throughout the food web. However, we note that additional processes may exhibit temperature-dependence including maturation (Jonsson et al., 2013), larval survivorship (McLeod et al., 2013), or rates governing dynamics of basal resources (Lindmark, Audzijonyte, et al., 2022). For the latter, in the context of one food web, temperaturedependencies were found to determine whether young age classes responded positively or negatively to warming in terms of size-at-age (Lindmark, Audzijonyte, et al., 2022). Further, an assessment of the total effect of warming on food webs should also address changes in productivity at the base of the food web resulting from, for instance, stronger stratification (Tittensor et al., 2021). Our comparative approach could be adapted to characterize the strength of these and other potential temperature dependence relationships across a variety of food webs. For instance, while we focused our analysis on exponential scalings with temperature, unimodal relationships are also feasible over larger temperature ranges with respect to rates governing feeding (Englund et al., 2011; Portner & Peck, 2010; Rall et al., 2012). Methods for representing and parameterizing unimodal relationships in food webs have been proposed (e.g., Delong et al., 2017; Woodworth-Jefcoats et al., 2019) and could be included in future comparative studies or as part of a model ensemble for a given system (e.g., Reum et al., 2020).

Our findings demonstrate the important role temperature dependence assumptions have on food web responses to warming and indicate the additional uncertainty they may impart into climate-forced projections. Ensemble approaches may help represent this uncertainty (Cheung et al., 2016; Hill et al., 2007) and we encourage continued development of methods to statistically weight ensemble member projections according to skill (e.g., Chandler, 2013; Spence et al., 2018) along with other approaches to better bound and communicate projection uncertainty (Parker, 2013). More generally, our results suggest that past efforts to generate ensemble projections using simplified, generic global food web models (e.g., Bryndum-Buchholz et al., 2020; Lotze et al., 2019; Tittensor et al., 2021) likely underestimate projection uncertainty when alternative temperature-dependence assumptions are not considered in the model set. That said, the models in our study, which explore warming effects on a narrow set of rates, are relatively consistent in projecting reductions in total spawning biomass on average under warming scenarios. This agrees directionally with global ensemble projections consisting of several types of models that differ with respect to structure and forcing variables, and that integrate warming impacts on food availability at the base of the food web (Lotze et al., 2019; Tittensor et al., 2021). While we have focused our analysis on temperature, theoretical frameworks for incorporating oxygen- and CO<sub>2</sub>-dependencies on biological rates are also developing (Neubauer & Andersen, 2019; Portner & Knust, 2007; Portner & Peck, 2010), and future studies are needed to evaluate how additional climate-linked changes in marine environments modify food web outcomes under warming.

### **Data Availability Statement**

All simulation outputs analyzed in the present study, along with R code which updates mizer model parameters according to specified temperatures, are available in an online repository [Software] (Reum, 2023, https://fig-share.com/s/7ce66c97db8beca8c1c3). Code for the trait-based models and the calibrated North Sea model are available as part of the R package "mizer" (https://cran.r-project.org/web/packages/mizer/). Calibrated regional models are available for: the Eastern Bering Sea [Software] (Reum, 2018); Coastal Tasmania [Software] (Audzijonyte, 2022); SE Australia [Software] (Novaglio, 2021); Central North Pacific [Software] (Woodworth-Jefcoats, 2020); and Baltic Sea [Software] (Delius et a. 2022).

### References

Al-Habsi, S. H., Sweeting, C. J., Polunin, N. V. C., & Graham, N. A. J. (2008). δ<sup>15</sup>N and δ<sup>13</sup>C elucidation of size-structured food webs in a Western Arabian Sea demersal trawl assemblage. *Marine Ecology Progress Series*, 353, 55–63. https://doi.org/10.3354/meps07167

Andersen, K. H. (2019). Fish ecology, evolution, and exploitation: A new theoretical synthesis. In *Fish ecology, evolution, and exploitation*. Princeton University Press. https://doi.org/10.1515/9780691189260

Andersen, K. H., & Pedersen, M. (2010). Damped trophic cascades driven by fishing in model marine ecosystems. Proceedings of the Royal Society B: Biological Sciences, 277(1682), 795–802. https://doi.org/10.1098/rspb.2009.1512

Audzijonyte, A. (2022). Size based mizer model for coastal Tasmania with multiple primary production pathways (version 1.0.0) [Software]. *Github*. https://github.com/astaaudzi/SEAmodel/releases/tag/v1.0.0

### Acknowledgments

JCPR and PWJ were supported in part through funding provided by a NOAA Fisheries International Science Fellowship. We also acknowledge support from the Australian Research Council Discovery Project DP170104240 ("Rewiring Marine Foodwebs") and August T. Larsson Guest Research Programme at SLU. We thank A. Whitehouse and I. Kaplan for comments on earlier versions of the manuscript. The findings and conclusions in this article are those of the authors and do not necessarily represent the views of the National Marine Fisheries Service, NOAA. Reference to trade names does not imply endorsement by the National Marine Fisheries Service, NOAA

Andersen, K. H., Jacobsen, N. S., & Farnsworth, K. D. (2016). The theoretical foundations for size spectrum models of fish communities. Canadian Journal of Fisheries and Aquatic Sciences, 73(4), 575–588. https://doi.org/10.1139/cjfas-2015-0230

- Audzijonyte, A., Delius, G. W., Stuart-Smith, R., Novaglio, C., Edgar, G., Barrett, N. S., & Blanchard, J. L. (2023). Changes in sea floor productivity are crucial to understanding the impact of climate change in temperate coastal ecosystems according to a new size-based model. *PLoS Biology*, 21(12), e3002392. https://doi.org/10.1371/journal.pbio.3002392
- Beamish, F. W. H. (1964). Respiration of fishes with special emphasis on standard oxygen consumption II. Influence of weight and temperature on respiration of several species. *Canadian Journal of Zoology*, 42(2), 177–188. https://doi.org/10.1139/z64-016
- Bindoff, N. L., Cheung, W. W. L., Kairo, J. G., Arístegui, J., Guinder, V. A., Hallberg, R., et al. (2019). Changing ocean, marine ecosystems, and dependent communities. In H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. (Eds.), *IPCC special* report on the ocean and cryosphere in a changing climate (pp. 447–587). Cambridge University Press. https://doi.org/10.1017/ 9781009157964.007
- Blanchard, J. L., Andersen, K. H., Scott, F., Hintzen, N. T., Piet, G., & Jennings, S. (2014). Evaluating targets and trade-offs among fisheries and conservation objectives using a multispecies size spectrum model. *Journal of Applied Ecology*, 51(3), 612–622. https://doi.org/10.1111/1365-2664.12238
- Blanchard, J. L., Heneghan, R. F., Everett, J. D., Trebilco, R., & Richardson, A. J. (2017). From bacteria to whales: Using functional size spectra to model marine ecosystems. *Trends in Ecology & Evolution*, 32(3), 174–186. https://doi.org/10.1016/j.tree.2016.12.003
- Blanchard, J. L., Jennings, S., Holmes, R., Harle, J., Merino, G., Allen, J. I., et al. (2012). Potential consequences of climate change for primary production and fish production in large marine ecosystems. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 367(1605), 2979–2989. https://doi.org/10.1098/rstb.2012.0231
- Blanchard, J. L., Jennings, S., Law, R., Castle, M. D., McCloghrie, P., Rochet, M., & Benoît, E. (2009). How does abundance scale with body size in coupled size-structured food webs? *Journal of Animal Ecology*, 78(1), 270–280. https://doi.org/10.1111/j.1365-2656.2008.01466.x
- Brown, J. H., Gillooly, J. F., Allen, A. P., Savage, V. M., & West, G. B. (2004). Toward a metabolic theory of ecology. *Ecology*, 85(7), 1771–1789. https://doi.org/10.1890/03-9000
- Bryndum-Buchholz, A., Prentice, F., Tittensor, D. P., Blanchard, J. L., Cheung, W. W. L., Christensen, V., et al. (2020). Differing marine animal biomass shifts under 21st century climate change between Canada's three oceans. *Facets*, 5(1), 105–122. https://doi.org/10.1139/facets-2019-0035
- Chandler, R. E. (2013). Exploiting strength, discounting weakness: Combining information from multiple climate simulators. *Philosophical Transactions of the Royal Society A: Mathematical, Physical & Engineering Sciences*, 371(1991), 20120388. https://doi.org/10.1098/rsta.2012.0388
- Cheung, W. W. L., Frölicher, T. L., Asch, R. G., Jones, M. C., Pinsky, M. L., Reygondeau, G., et al. (2016). Building confidence in projections of the responses of living marine resources to climate change. *ICES Journal of Marine Science*, 73(5), 1283–1296. https://doi.org/10.1093/ icesjms/fsv250
- Coghlan, A. R., Blanchard, J. L., Heather, F. J., Stuart-Smith, R. D., Edgar, G. J., & Audzijonyte, A. (2022). Community size structure varies with predator–prey size relationships and temperature across Australian reefs. *Ecology and Evolution*, 12(4), e8789. https://doi.org/10.1002/ece3. 8789
- Delius, G. W., Scott, F., Southwell, R., Audzijonyte, A., Lindmark, M., & Blanchard, J. (2022). maxlindmark/mizer-rewiring: v1\_baltic\_mizer (v1\_baltic\_mizer) [Software]. Zenodo. https://doi.org/10.5281/zenodo.6821926
- DeLong, J. P., Gibert, J. P., Luhring, T. M., Bachman, G., Reed, B., Neyer, A., & Montooth, K. L. (2017). The combined effects of reactant kinetics and enzyme stability explain the temperature dependence of metabolic rates. *Ecology and Evolution*, 7(11), 3940–3950. https://doi.org/ 10.1002/ece3.2955
- Doney, S. C., Ruckelshaus, M. J., Duffy, E., Barry, J. P., Chan, F., English, C. A., et al. (2012). Climate change impacts on marine ecosystems. Annual Review of Marine Science, 4(1), 11–37. https://doi.org/10.1146/annurev-marine-041911-111611
- Englund, G., Öhlund, G., Hein, C. L., & Diehl, S. (2011). Temperature dependence of the functional response. *Ecology Letters*, 14(9), 914–921. https://doi.org/10.1111/j.1461-0248.2011.01661.x
- García, B. G., Valverde, J. C., Aguado-Giménez, F., García, J. G., & Hernández, M. D. (2011). Effect of the interaction between body weight and temperature on growth and maximum daily food intake in sharpsnout sea bream (*Diplodus puntazzo*). Aquaculture International, 19(1), 131–141. https://doi.org/10.1007/s10499-010-9347-2
- Gårdmark, A., & Huss, M. (2020). Individual variation and interactions explain food web responses to global warming. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375(1814), 20190449. https://doi.org/10.1098/rstb.2019.0449
- Gårdmark, A., Lindegren, M., Neuenfeldt, S., Blenckner, T., Heikinheimo, O., Müller-Karulis, B., et al. (2013). Biological ensemble modeling to evaluate potential futures of living marine resources. *Ecological Applications*, 23(4), 742–754. https://doi.org/10.1890/12-0267.1
- Gibert, J. P. (2019). Temperature directly and indirectly influences food web structure. Scientific Reports, 9(1), 5312. https://doi.org/10.1038/ s41598-019-41783-0
- Gillooly, J. F., Brown, J. H., West, G. B., Savage, V. M., & Charnov, E. L. (2001). Effects of size and temperature on metabolic rate. *Science*, 293(5538), 2248–2251. https://doi.org/10.1126/science.1061967
- Gilman, S. E., Urban, M. C., Tewksbury, J., Gilchrist, G. W., & Holt, R. D. (2010). A framework for community interactions under climate change. Trends in Ecology & Evolution, 25(6), 325–331. https://doi.org/10.1016/j.tree.2010.03.002
- Hartvig, M., Andersen, K. H., & Beyer, J. E. (2011). Food web framework for size-structured populations. *Journal of Theoretical Biology*, 272(1), 113–122. https://doi.org/10.1016/j.jtbi.2010.12.006
- Heneghan, R. F., Galbraith, E., Blanchard, J. L., Harrison, C., Barrier, N., Bulman, C., et al. (2021). Disentangling diverse responses to climate change among global marine ecosystem models. *Progress in Oceanography*, 198, 102659. https://doi.org/10.1016/j.pocean.2021.102659
- Heneghan, R. F., Hatton, I. A., & Galbraith, E. D. (2019). Climate change impacts on marine ecosystems through the lens of the size spectrum. *Emerging Topics in Life Sciences*, 3(2), 233–243. https://doi.org/10.1042/etls20190042
- Hill, S. L., Watters, G. M., Punt, A. E., McAllister, M. K., Quéré, C. L., & Turner, J. (2007). Model uncertainty in the ecosystem approach to fisheries. *Fish and Fisheries*, 8(4), 315–336. https://doi.org/10.1111/j.1467-2979.2007.00257.x
- Horn, S., Meunier, C. L., Fofonova, V., Wiltshire, K. H., Sarker, S., Pogoda, B., & Asmus, H. (2021). Toward improved model capacities for assessment of climate impacts on coastal bentho-pelagic food webs and ecosystem services. *Frontiers in Marine Science*, 8, 819. https://doi. org/10.3389/fmars.2021.567266
- Jacobsen, N. S., Burgess, M. G., & Andersen, K. H. (2017). Efficiency of fisheries is increasing at the ecosystem level. Fish and Fisheries, 18(2), 199–211. https://doi.org/10.1111/faf.12171
- Jennings, S., & Blanchard, J. L. (2004). Fish abundance with no fishing: Predictions based on macroecological theory. *Journal of Animal Ecology*, 73(4), 632–642. https://doi.org/10.1111/j.0021-8790.2004.00839.x

- Jennings, S., Mélin, F., Blanchard, J. L., Forster, R. M., Dulvy, N. K., & Wilson, R. W. (2008). Global-scale predictions of community and ecosystem properties from simple ecological theory. *Proceedings of the Royal Society B: Biological Sciences*, 275(1641), 1375–1383. https://doi.org/10.1098/rspb.2008.0192
- Jonsson, N., & Finstad, A. G. (2013). Effects of temperature and food quality on age and size at maturity in ectotherms: An experimental test with Atlantic salmon. *Journal of Animal Ecology*, 82(1), 201–210. https://doi.org/10.1111/j.1365-2656.2012.02022.x
- Keil, G., Cummings, E., & de Magalhaes, J. P. (2015). Being cool: How body temperature influences ageing and longevity. *Biogerontology*, 16(4), 383–397. https://doi.org/10.1007/s10522-015-9571-2

Kerr, S. R., & Dickie, L. M. (2001). The biomass spectrum. Columbia University Press.

- Killen, S. S., Atkinson, D., & Glazier, D. S. (2010). The intraspecific scaling of metabolic rate with body mass in fishes depends on lifestyle and temperature. *Ecology Letters*, 13(2), 184–193. https://doi.org/10.1111/j.1461-0248.2009.01415.x
- Kooijman (2000). Dynamic energy and mass budgets in biological systems (2nd ed., p. 426). Cambridge University Press.
- Kwiatkowski, L., Aumont, O., & Bopp, L. (2019). Consistent trophic amplification of marine biomass declines under climate change. Global Change Biology, 25(1), 219–229. https://doi.org/10.1111/gcb.14468
- Kwiatkowski, L., Torres, O., Bopp, L., Aumont, O., Chamberlain, M., Christian, J. R., et al. (2020). Twenty-first century ocean warming, acidification, deoxygenation, and upper-ocean nutrient and primary production decline from CMIP6 model projections. *Biogeosciences*, 17(13), 3439–3470. https://doi.org/10.5194/bg-17-3439-2020
- Law, R., Plank, M. J., James, A., & Blanchard, J. L. (2009). Size-spectra dynamics from stochastic predation and growth of individuals. *Ecology*, 90(3), 802–811. https://doi.org/10.1890/07-1900.1

Lefort, S., Aumont, O., Bopp, L., Arsouze, T., Gehlen, M., & Maury, O. (2015). Spatial and body-size dependent response of marine pelagic communities to projected global climate change. *Global Change Biology*, 21(1), 154–164. https://doi.org/10.1111/gcb.12679 Legendre, P., & Legendre, L. (2012). *Numerical ecology*, Elsevier.

Lemoine, N. P., & Burkepile, D. E. (2012). Temperature-induced mismatches between consumption and metabolism reduce consumer fitness. *Ecology*, 93(11), 2483–2489. https://doi.org/10.1890/12-0375.1

- Lindmark, M., Audzijonyte, A., Blanchard, J. L., & Gårdmark, A. (2022). Temperature impacts on fish physiology and resource abundance lead to faster growth but smaller fish sizes and yields under warming. *Global Change Biology*, 28(21), 6239–6253. https://doi.org/10.1111/gcb.16341
- Lindmark, M., Huss, M., Ohlberger, J., & G\aardmark, A. (2018). Temperature-dependent body size effects determine population responses to climate warming. *Ecology Letters*, 21(2), 181–189. https://doi.org/10.1111/ele.12880
- Lindmark, M., Ohlberger, J., & Gårdmark, A. (2022). Optimum growth temperature declines with body size within fish species. Global Change Biology, 28(7), 2259–2271. https://doi.org/10.1111/gcb.16067

Lindmark, M., Ohlberger, J., Huss, M., & Gårdmark, A. (2019). Size-based ecological interactions drive food web responses to climate warming. *Ecology Letters*, 22(5), 778–786. https://doi.org/10.1111/ele.13235

- Lord, J. P., Barry, J. P., & Graves, D. (2017). Impact of climate change on direct and indirect species interactions. *Marine Ecology Progress Series*, 571, 1–11. https://doi.org/10.3354/meps12148
- Lotze, H. K., Tittensor, D. P., Bryndum-Buchholz, A., Eddy, T. D., Cheung, W. W., Galbraith, E. D., et al. (2019). Global ensemble projections reveal trophic amplification of ocean biomass declines with climate change. *Proceedings of the National Academy of Sciences of the United States of America*, 116(26), 12907–12912. https://doi.org/10.1073/pnas.1900194116
- Maury, O., Faugeras, B., Shin, Y.-J., Poggiale, J.-C., Ari, T. B., & Marsac, F. (2007). Modeling environmental effects on the size-structured energy flow through marine ecosystems. Part 1: The model. *Progress in Oceanography*, 74(4), 479–499. https://doi.org/10.1016/j.pocean. 2007.05.002
- Maury, O., Shin, Y.-J., Faugeras, B., Ari, T. B., & Marsac, F. (2007). Modeling environmental effects on the size-structured energy flow through marine ecosystems. Part 2: Simulations. Progress in Oceanography, 74(4), 500–514. https://doi.org/10.1016/j.pocean.2007.05.001

McLeod, I. M., Rummer, J. L., Clark, T. D., Jones, G. P., McCormick, M. I., Wenger, A. S., & Munday, P. L. (2013). Climate change and the performance of larval coral reef fishes: The interaction between temperature and food availability. *Conservation Physiology*, 1(1), cot024. https://doi.org/10.1093/conphys/cot024

- Messmer, V., Pratchett, M. S., Hoey, A. S., Tobin, A. J., Coker, D. J., Cooke, S. J., & Clark, T. D. (2017). Global warming may disproportionately affect larger adults in a predatory coral reef fish. *Global Change Biology*, 23(6), 2230–2240. https://doi.org/10.1111/gcb.13552
- Munch, S. B., & Salinas, S. (2009). Latitudinal variation in lifespan within species is explained by the metabolic theory of ecology. Proceedings of the National Academy of Sciences of the United States of America, 106(33), 13860–13864. https://doi.org/10.1073/pnas.0900300106

Neubauer, P., & Andersen, K. H. (2019). Thermal performance of fish is explained by an interplay between physiology, behaviour and ecology. *Conservation Physiology*, 7(1), coz025. https://doi.org/10.1093/conphys/coz025

Novaglio, C. (2021). camillanovaglio/mixed\_fisheries\_tradeoff (version 1.0.0) [Software]. Zenodo. https://doi.org/10.5281/zenodo.5715717

- Novaglio, C., Blanchard, J. L., Plank, M. J., van Putten, E. I., Audzijonyte, A., Porobic, J., & Fulton, E. A. (2022). Exploring trade-offs in mixed fisheries by integrating fleet dynamics into multispecies size-spectrum models. *Journal of Applied Ecology*, 59(3), 715–728. https://doi.org/10. 1111/1365-2664.14086
- Ohlberger, J., Edeline, E., Vøllestad, L. A., Stenseth, N. C., & Claessen, D. (2011). Temperature-driven regime shifts in the dynamics of sizestructured populations. *The American Naturalist*, 177(2), 211–223. https://doi.org/10.1086/657925
- Ohlberger, J., Mehner, T., Staaks, G., & Hölker, F. (2012). Intraspecific temperature dependence of the scaling of metabolic rate with body mass in fishes and its ecological implications. *Oikos*, 121(2), 245–251. https://doi.org/10.1111/j.1600-0706.2011.19882.x
- Parker, W. S. (2013). Ensemble modeling, uncertainty and robust predictions. Wiley Interdisciplinary Reviews: Climate Change, 4(3), 213–223. https://doi.org/10.1002/wcc.220
- Pauly, D. (1980). On the interrelationships between natural mortality, growth parameters, and mean environmental temperature in 175 fish stocks. *ICES Journal of Marine Science*, 39(2), 175–192. https://doi.org/10.1093/icesjms/39.2.175
- Peters, R. H. (1983). The ecological implications of body size. Cambridge University Press.
- Petrik, C. M., Stock, C. A., Andersen, K. H., van Denderen, P. D., & Watson, J. R. (2019). Bottom-up drivers of global patterns of demersal, forage, and pelagic fishes. *Progress in Oceanography*, 176, 102124. https://doi.org/10.1016/j.pocean.2019.102124
- Portner, H. O., & Knust, R. (2007). Climate change affects marine fishes through the oxygen limitation of thermal tolerance. *Science*, 315(5808), 95–97. https://doi.org/10.1126/science.1135471
- Portner, H. O., & Peck, M. A. (2010). Climate change effects on fishes and fisheries: Towards a cause-and-effect understanding. *Journal of Fish Biology*, 77(8), 1745–1779. https://doi.org/10.1111/j.1095-8649.2010.02783.x
- Rall, B. C., Brose, U., Hartvig, M., Kalinkat, G., Schwarzmüller, F., Vucic-Pestic, O., & Petchey, O. L. (2012). Universal temperature and bodymass scaling of feeding rates. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 367(1605), 2923–2934. https://doi.org/ 10.1098/rstb.2012.0242

- Reum, J. C., Jennings, S., & Hunsicker, M. E. (2015). Implications of scaled  $\delta^{15}$ N fractionation for community predator–prey body mass ratio estimates in size-structured food webs. *Journal of Animal Ecology*, 84(6), 1618–1627. https://doi.org/10.1111/1365-2656.12405
- Reum, J. C. P. (2018). Data from: Species-specific ontogenetic diet shifts attenuate trophic cascades and lengthen food chains in an exploited ecosystem (version 1.0.0) [Software]. Figshare. https://doi.org/10.6084/m9.figshare.7158635.v1
- Reum, J. C. P. (2023). Outputs and supplemental code for simulated warming of size spectrum food webs [Software]. *FigShare*. https://figshare. com/s/7ce66c97db8beca8c1c3
- Reum, J. C. P., Blanchard, J. L., Holsman, K. K., Aydin, K., Hollowed, A. B., Hermann, A. J., et al. (2020). Ensemble projections of future climate change impacts on the Eastern Bering Sea food web using a multispecies size spectrum model. *Frontiers in Marine Science*, 7. https://doi.org/ 10.3389/fmars.2020.00124
- Reum, J. C. P., Blanchard, J. L., Holsman, K. K., Aydin, K., & Punt, A. E. (2019). Species-specific ontogenetic diet shifts attenuate trophic cascades and lengthen food chains in an exploited ecosystem. *Oikos*, 128(7), 1051–1064. https://doi.org/10.1111/oik.05630
- Reum, J. C. P., Holsman, K. K., Aydin, K. Y., Blanchard, J. L., & Jennings, S. (2019). Energetically relevant predator–prey body mass ratios and their relationship with predator body size. *Ecology and Evolution*, 9(1), 201–211. https://doi.org/10.1002/ece3.4715
- Savage, V. M., Gillooly, J. F., Brown, J. H., West, G. B., & Charnov, E. L. (2004). Effects of body size and temperature on population growth. *The American Naturalist*, 163(3), 429–441. https://doi.org/10.1086/381872
- Scott, F., Blanchard, J. L., & Andersen, K. H. (2014). mizer: An R package for multispecies, trait-based and community size spectrum ecological modelling. *Methods in Ecology and Evolution*, 5(10), 1121–1125. https://doi.org/10.1111/2041-210X.12256
- Sentis, A., Hemptinne, J.-L., & Brodeur, J. (2014). Towards a mechanistic understanding of temperature and enrichment effects on species interaction strength, omnivory and food-web structure. *Ecology Letters*, 17(7), 785–793. https://doi.org/10.1111/ele.12281
- Sheldon, R. W., Prakash, A., & Sutcliffe, W. H. (1972). Size distribution of particles in the ocean. *Limnology & Oceanography*, 17(3), 327–340. https://doi.org/10.4319/10.1972.17.3.0327
- Spence, M. A., Blanchard, J. L., Rossberg, A. G., Heath, M. R., Heymans, J. J., Mackinson, S., et al. (2018). A general framework for combining ecosystem models. *Fish and Fisheries*, 19(6), 1031–1042. https://doi.org/10.1111/faf.12310
- Strong, K. W., & Daborn, G. R. (1980). The influence of temperature on energy budget variables, body size, and seasonal occurrence of the isopod Idotea baltica (Pallas). Canadian Journal of Zoology, 58(11), 1992–1996. https://doi.org/10.1139/z80-274
- Tekwa, E. W., Watson, J. R., & Pinsky, M. L. (2022). Body size and food-web interactions mediate species range shifts under warming. Proceedings of the Royal Society B: Biological Sciences, 289(1972), 20212755. https://doi.org/10.1098/rspb.2021.2755
- Thorson, J. T., Munch, S. B., Cope, J. M., & Gao, J. (2017). Predicting life history parameters for all fishes worldwide. *Ecological Applications*, 27(8), 2262–2276. https://doi.org/10.1002/eap.1606
- Thunell, V., Lindmark, M., Huss, M., & Gårdmark, A. (2021). Effects of warming on intraguild predator communities with ontogenetic diet shifts. *The American Naturalist*, 198(6), 706–718. https://doi.org/10.1086/716927
- Tittensor, D. P., Novaglio, C., Harrison, C. S., Heneghan, R. F., Barrier, N., Bianchi, D., et al. (2021). Next-generation ensemble projections reveal higher climate risks for marine ecosystems. *Nature Climate Change*, 11(11), 973–981. https://doi.org/10.1038/s41558-021-01173-9
- Trebilco, R., Baum, J. K., Salomon, A. K., & Dulvy, N. K. (2013). Ecosystem ecology: Size-based constraints on the pyramids of life. Trends in Ecology & Evolution, 28(7), 423–431. https://doi.org/10.1016/j.tree.2013.03.008
- Trebilco, R., Dulvy, N. K., Anderson, S. C., & Salomon, A. K. (2016). The paradox of inverted biomass pyramids in kelp forest fish communities. Proceedings of the Royal Society B: Biological Sciences, 283(1833), 20160816. https://doi.org/10.1098/rspb.2016.0816
- Vucic-Pestic, O., Ehnes, R. B., Rall, B. C., & Brose, U. (2011). Warming up the system: Higher predator feeding rates but lower energetic efficiencies. *Global Change Biology*, 17(3), 1301–1310. https://doi.org/10.1111/j.1365-2486.2010.02329.x
- Woodworth-Jefcoats, P. A. (2020). pwoodworth-jefcoats/size-based-modeling [Software]. Github. https://github.com/pwoodworth-jefcoats/Size-Based-Modeling
- Woodworth-Jefcoats, P. A., Blanchard, J. L., & Drazen, J. C. (2019). Relative impacts of simultaneous stressors on a pelagic marine ecosystem. Frontiers in Marine Science, 6, 383. https://doi.org/10.3389/fmars.2019.00383
- Woodworth-Jefcoats, P. A., Polovina, J. J., Dunne, J. P., & Blanchard, J. L. (2013). Ecosystem size structure response to 21st century climate projection: Large fish abundance decreases in the central North Pacific and increases in the California Current. *Global Change Biology*, 19(3), 724–733. https://doi.org/10.1111/gcb.12076
- Wootton, H. F., Morrongiello, J. R., Schmitt, T., & Audzijonyte, A. (2022). Smaller adult fish size in warmer water is not explained by elevated metabolism. *Ecology Letters*, 25(5), 1177–1188. https://doi.org/10.1111/ele.13989
- Xie, X., & Sun, R. (1990). The bioenergetics of the Southern Catfish (Silurus meridionalis Chen). I. Resting metabolic rate as a function of body weight and temperature. *Physiological Zoology*, 63(6), 1181–1195. https://doi.org/10.1086/physzool.63.6.30152639
- Zhang, L., Takahashi, D., Hartvig, M., & Andersen, K. H. (2017). Food-web dynamics under climate change. Proceedings of the Royal Society B: Biological Sciences, 284(1867), 20171772. https://doi.org/10.1098/rspb.2017.1772